Author's response to reviews

Title: Smoking, alcohol consumption, physical activity, and family history and the risks of acute myocardial infarction and unstable angina pectoris: a prospective cohort study

Authors:

Audrey HH Merry (Audrey.Merry@epid.unimaas.nl)
Jolanda MA Boer (Jolanda.Boer@rivm.nl)
Leo J Schouten (LJ.Schouten@epid.unimaas.nl)
Edith JM Feskens (Edith.Feskens@wur.nl)
WM Monique Verschuren (Monique.Verschuren@rivm.nl)
Anton PM Gorgels (t.gorgels@mumc.nl)
Piet A van den Brandt (PA.vandenBrandt@epid.unimaas.nl)

Version: 5 Date: 24 November 2010

Author's response to reviews: see over
Dear Editor,

Thank you for the opportunity to check the edited version of our manuscript ‘Smoking, alcohol consumption, physical activity, and family history and the risks of acute myocardial infarction and unstable angina pectoris: a prospective cohort study’.

In the manuscript, we made the following changes:

Major revisions:
- All highlighting and tracking have been removed from the manuscript.

Minor revisions:
- The abbreviations are now included as a paragraph instead of as a list.
- Some typing errors are corrected.

We hope that you are satisfied with the corrections and that this manuscript is now acceptable for publication in your journal.

Yours sincerely,

Audrey H.H. Merry, MSc
Department of Epidemiology,
Maastricht University
The Netherlands